Author's response to reviews

Title: PROspective MEmory Training to improve HEart failUre Self-care (PROMETHEUS): study protocol for a randomised controlled trial

Authors:

Jan Cameron (jan.cameron@acu.edu.au)
Peter G Rendell (peter.rendell@acu.edu.au)
Chantal F Ski (chantal.ski@acu.edu.au)
Christina E Kure (christina.kure@acu.edu.au)
Skye S McLennan (skye.mclennan@acu.edu.au)
Nathan S Rose (Nathan.Rose@acu.edu.au)
David L Prior (david.prior@svhm.org.au)
David R Thompson (david.thompson@acu.edu.au)

Version: 4
Date: 13 April 2015

Author's response to reviews:

The following comments from the reviewer have been addressed and are highlighted in the paper:

• Trial status should be update
Response: this has been updated to 'This study is actively recruiting and ongoing'.
• The figure might show the matched healthy control group in addition to the randomized groups
Response: we have added in the figure a box to indicate the matched healthy control group
• Under Author's Contributions, "PG" should probably be "PGR" or "PR" and 'DR' should be 'DP'
Response: Thank you for highlighting this error which has been corrected to 'PR' and 'DP'